Author's response to reviews

Title: Subjective health complaints and reduced general function are not associated with tick bites or antibodies to Borrelia burgdorferi sens u lato in blood donors in western Norway

Authors:

Reidar Hjetland (reidar.hjetland@helse-forde.no)
Harald Reiso (harald.reiso@medisin.uio.no)
Camilla Ihlebæk (camilla.ihlebak@nmbu.no)
Roy M Nilsen (roy.miodini.nilsen@helse-bergen.no)
Nils Grude (nigrud@siv.no)
Elling Ulvestad (elling.ulvestad@helse-bergen.no)

Version: 2
Date: 10 April 2015

Author's response to reviews:

The Editor,

Required revisions pointed out in your email of April 9th (full postal address of submitting author, and adding sections on Competing interests and Authors' contributions) have been corrected. I hope that you will find it satisfactory.

Yours sincerely,
Reidar Hjetland